Author's response to reviews

Title: DALI: Vitamin D And lifestyle intervention for gestational diabetes mellitus (GDM) prevention: an European multicentre, randomised trial - study protocol

Authors:

Judith G.M. Jelsma (j.jelsma@vumc.nl)
Mireille M.N. van Poppel (Mnm.vanpoppel@vumc.nl)
Sander Galjaard (sander.galjaard@uzleuven.be)
Gernot Desoye (gernot.desoye@medunigraz.at)
Rosa Corcoy (RCorcoy@santpau.cat)
Roland Devlieger (roland.devlieger@uzleuven.be)
Andre van Assche (andre.vanassche@med.kuleuven.be)
Dirk Timmerman (dirk.timmerman@uz.kuleuven.ac.be)
Jürgen Harreiter (juergen.harreiter@meduniwien.ac.at)
Alexandra Kautzky-Willer (alexandra.kautzky-willer@meduniwien.ac.at)
Peter Damm (pdamm@dadlnet.dk)
Elisabeth R. Mathiesen (Elisabeth.Mathiesen@regionh.dk)
Dorte M. Jensen (Dorte.Moeller.Jensen@ouh.regionsyddanmark.dk)
Liselotte Andersen (ll.torvin@dadlnet.dk)
Fidelma Dunne (fidelma.dunne@nuigalway.ie)
Annunziata Lapolla (annunziata.lapolla@unipd.it)
Graziano Di Cianni (graziano.dicianni@gmail.com)
Alessandra Bertolotto (alessandrabertolotto1959@yahoo.it)
Ewa Wender-Oegowska (ewaoz@post.home.pl)
Agnieszka Zawiejska (agazaw@post.home.pl)
Kinga Blumska (kinga.blumska@gmail.com)
David Hill (David.hill@lawsonresearch.com)
Pablo Rebollo (pablo@baphealth.com)
Frank J. Snoek (F.J.snoek@vumc.nl)
David Simmons (david.simmons@addenbrookes.nhs.uk)

Version: 2 Date: 4 June 2013

Author's response to reviews: see over
Dear Editor,

Hereby we would like to submit the revised study protocol "DALI: Vitamin D And lifestyle intervention for gestational diabetes mellitus (GDM) prevention: an European multicentre, randomised trial – study protocol" for publication in BMC Pregnancy and Childbirth.

The following changes have been made since the last version, which was submitted on 28-5-2013.

1. The grant number from the Netherlands Organization for Scientific Research has been added to the acknowledgement section. The number is: 036.002.408

2. The DALI study has received external funding: European Community's 7th Framework Programme (FP7/2007-2013) under grant agreement no 242187
   a. The document has been send by email to you

3. The DALI study has received ethical approval in all the participating countries.
   a. Ethical approval from the main ethical committees in the Netherlands, Austria, Poland and Spain has been send by email to you.

4. Currently no other manuscripts are under submission with other journals.

We hope that we provided all the missing information.

Best regards,

Mireille NM van Poppel